SYLLABUS: MATH 555, 704I REAL ANALYSIS II

Instructor: Michael Filaseta

Office: 301 LeConte

Office Phone Number: 777-6589

Office Hours: By Appointment (WF mornings and afternoons, TTh afternoons)

Textbook (optional): Introduction to Real Analysis by Manfred Stoll

Grading Percentages: Homework, Quizzes, and Class Participation (25% total) 2 tests (20% each) final (35%)

Grading Scale: A score of 90% or above will be an A a score of at least 87 % will be at least a B+ a score of at least 80 % will be at least a B a score of at least 77 % will be at least a C+ a score of at least 70 % will be at least a C a score of at least 67 % will be at least a D+ a score of at least 60 % will be at least a D a score of less than 60 % will be an F

Additional Comments:

(1) No calculators are allowed on tests or the final exam.

(2) You should feel free to work on homework together; you should share ideas but are NOT to share answers. Your solutions to homework problems should be written up individually (and not copied).